Author's response to reviews

Title: Factors influencing the decision that women make on their mode of delivery: the Health Belief Model.

Authors:

Alice Yuen Loke (hsaloke@polyu.edu.hk)
Louise Davis (Glennandlou@gmail.com)
Sau-fun Li (sau.fun/li@connect.polyu.hk)

Version: 2
Date: 28 July 2014

Author's response to reviews:

1. Title page: More detail information containing the names institutions, countries and email addresses of all authors, and the full postal address of the submitting author were added to the title page.

2. Email address: hsaloke@polyu.edu.hk is the correct email address.

3. Please follow the correct format for your Authors' Contributions section: The format of this section is changed accordingly.

4. Line numbering: added to the resubmission.